Author's response to reviews

Title: The State of the Research for Health Environment in the Ministries of Health of the Economic Community of the West African States (ECOWAS)

Authors:

Issiaka Sombie (isombie@wahooas.org)
Jude Aidam (jaidam@wahooas.org)
Blahima Konaté (kobla70@hotmail.com)
Télesphore D Somé (telesome@hotmail.com)
Stanislas S Kambou (skambou@wahooas.org)

Version: 2 Date: 9 August 2013

Author's response to reviews:

We want to express our sincere thanks to the editorial team and especially the reviewers (Drs. RM and MM) for taking their precious time in review this work to get it to a publishable standard.

We appreciate every one of their comment and we have taken the time to do a thorough reassessment and the necessary corrections to address their concerns where we have the information. The detail to which they have gone in doing the review has been very helpful and have cause the author to represent the details of almost every section of this write-up to provide clarity of the title, concept, methods, results and discussions while maintain the integrity of the data already collected.

This review has been an important baseline for us in our activities and we hope to be able to share this information and learn more from the feedback given. There has been no attempt to over analyse the data received as the initial intent was internal operational purposes.